

June 29, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: ERP revision.docx).

**Title:** Endoscopic retrograde pancreatography: when should we do it?

**Author:** Renáta Bor, László Madácsy, Anna Fábián, Attila Szepes, Zoltán Szepes

**Name of Journal:** *World Journal of Gastrointestinal Endoscopy*

**ESPS Manuscript NO:** 19675

The manuscript has been improved according to the suggestions of reviewers:

1. We have created a table which summarizes the main complications of ERCP.
2. Revision has been made according to the suggestions of the reviewer<sup>1</sup>
  - (1) We discussed the role of ERP for anomalous junction of pancreato-biliary ductal system.
  - (2) In the conclusion the indications of ERP has more clearly defined.

Thank you again for publishing our manuscript in the *World Journal of Gastrointestinal Endoscopy*.

Sincerely yours,



Renáta BOR, MD  
First Department of Medicine  
University of Szeged  
Szeged, Korányi 8-10  
H-1083-Hungary  
Fax: +36-62-545-198  
E-mail: bor.reni86@gmail.com